Author’s response to reviews

Title: How does community context influence coalitions in the formation stage? A multiple case study based on the Community Coalition Action Theory.

Authors:

Michelle C Kegler (mkegler@sph.emory.edu)
Jessica A Rigler (Rigler@azdhs.gov)
Sally Honeycutt (shoneyc@sph.emory.edu)

Version: 4 Date: 26 January 2010

Author’s response to reviews:

The revisions were minor. We added a sentence saying the research had been classified as exempt after review by the Emory University IRB. We also put the tables in portrait layout.